Author's response to reviews

Title: A heart failure self-management program for patients of all literacy levels: A randomized, controlled trial [ISRCTN11535170]

Authors:

Darren A DeWalt (dewaltd@med.unc.edu)
Robert M Malone (rmalone@med.unc.edu)
Mary E Bryant (bbryant@med.unc.edu)
Margaret C Kosnar (margaret_corley@med.unc.edu)
Kelley E Corr (kelley_corr@med.unc.edu)
Russell L Rothman (russell.rothman@vanderbilt.edu)
Carla A Sueta (paris@med.unc.edu)
Michael P Pignone (pignone@med.unc.edu)

Version: 2 Date: 27 September 2005

Author's response to reviews: see over
September 26, 2005

Editor  
BioMed Central Health Services Research  
Middlesex House  
34-42 Cleveland Street  
London W1T 4LB  
UK

Dear Editor:

The only change since the original submission on 9/9/2005 is the addition of the clinical trial registration number: ISRCTN11535170.

Sincerely,

Darren A. DeWalt, MD, MPH  
5039 Old Clinic Building, CB#7110  
Chapel Hill, NC 27599  
Phone: (919) 966-2276  
Fax: (919) 966-2274  
e-mail: dewaltd@med.unc.edu